Author's response to reviews

Title: Determinants of patient satisfaction in ambulatory oncology: a cross sectional study based on the OUT-PATSAT35 questionnaire

Authors:

Thanh Vân, France NGUYEN (tvfnguyen@chu-besancon.fr)
Jean-François BOSSET (jean-francois.bosset@univ-fcomte.fr)
Alain MONNIER (moma25@orange.fr)
Valérie PERRIN (vperrin@chbm.fr)
Cédric BAUMANN (cedric.baumann@medecine.uhp-nancy.fr)
Anne BREDART (anne.bredart@curie.net)
Mariette MERCIER (mariette.mercier@univ-fcomte.fr)

Version: 3 Date: 10 November 2011

Author's response to reviews:

TVFN participated to the coordination of the study, performed the statistical analysis, the interpretation of data and drafted the manuscript. JFB and AM participated in the conception and design of the study. JF and VP contributed to the acquisition of data. CB and AB revised the manuscript critically for important intellectual content. MM participated in the conception and coordination of the study, and helped to the statistical analysis and the interpretation of data. All authors read and approved the final manuscript.